Author's response to reviews

Title: Partners of people on ART - a New Evaluation of the Risks (The PARTNER study): Design and Methods

Authors:

Alison J Rodger (alison.rodger@ucl.ac.uk)
Matthew Weait (m.weait@bbk.ac.uk)
Tina Bruun (tbr@cphiv.dk)
Pietro Vernazza (pietro.vernazza@kssg.ch)
Simon Collins (simon.collins@i-base.org.uk)
Vicente Estrada (vestrada.hcsc@salud.madrid.org)
Jan Van Lunzen (v.lunzen@uke.uni-hamburg.de)
Giulio Maria Corbelli (giuliomcorbelli@hotmail.com)
Andrew Phillips (andrew.phillips@ucl.ac.uk)
Jens Lundgren (jdl@cphiv.dk)

Version: 2 Date: 11 April 2012

Author's response to reviews:

The following text has been added to the methods section of the manuscript:

"In the UK, the study was reviewed and approved by the North West London REC 2 Ethics Committee (EC reference number 10/H0720/55). Ethics approval was obtained in-country for all other European sites involved in the study"